Author's response to reviews

Title: Gender differences in sex life issues - a population-based study of migraine sufferers

Authors:

Markku Sumanen (markku.sumanen@uta.fi)
Ansa Ojanlatva (ansoja@utu.fi)
Anna Rantala (anna.rantala@fimnet.fi)
Lauri Sillanmaki (lahesi@utu.fi)
Kari Mattila (kari.j.mattila@uta.fi)

Version: 4 Date: 7 April 2008

Author's response to reviews:

We thank the Editor and the BioMed Central Editorial Production Team for your positive attitude concerning our manuscript ‘Gender differences in sex life issues – a population-based study of migraine sufferers’ MS: 6475933671654548.

We have now taken into consideration all the formatting changes requested and revised the manuscript.

1. The list of authors is written exactly as they are in the submission system.
2. Each affiliation is written in full. Each author is also linked to their corresponding affiliation.
3. All unnecessary capitals have been removed in the headings and subheadings.
4. Keywords have been removed.
5. Conclusion section has been inserted after Discussion.
6. Table sizes have been corrected.
7. Tables have now been numbered with standard numerals.